Author's response to reviews

Title: Incidence and outcome of newly-diagnosed tuberculosis infections in schizophrenics: a 12-year, nationwide, retrospective longitudinal study

Authors:

Shu-Chen Kuo (ludwigvantw@gmail.com)
Yung-Tai Chen (ytchen0117@gmail.com)
Szu-Yuan Li (syliv@vghtpe.gov.tw)
Yi-Tzu Lee (s851009@yahoo.com.tw)
Albert C. Yang (accyang@gmail.com)
Te-Li Chen (tecklayyy@gmail.com)
Chia-Jen Liu (chiajenliu@gmail.com)
Tzeng-Ji Chen (tjchen@vghtpe.gov.tw)
Ih-Jen Su (suihjen@nhri.org.tw)
Chang-Phone Fung (cpfung@vghtpe.gov.tw)

Version: 3 Date: 20 May 2013

Author's response to reviews:

Thank you for your recent submission of the revised manuscript. However, before we can proceed there is still an issue with Table 4. That states "Only factors with P value less than 0.1 in the univariate analysis were entered into the multivariate analysis" yet Charlson's score, diabetes, CPD, MI, Heart failure are all NOT in the model.

Sorry for the unclarity. Only factors with P value less than 0.1 in the univariate analysis were entered into the multivariate analysis and only factors with P value less then 0.05 was presented in the table.

We had added the explanation on the footnote of Table 4.